Author's response to reviews

Title: Combined determination of plasma MMP2, MMP9, and TIMP1 improves the non-invasive detection of transitional cell carcinoma of the bladder

Authors:

Andrea Staack (andrea.staack@charite.de)
Steffen Badendieck (badendieck@12move.de)
Dietmar Schnorr (dietmar.schnorr@charite.de)
Stefan Loening (stefan.loening@charite.de)
Klaus Jung (klaus.jung@charite.de)

Version: 8 Date: 7 August 2006

Author's response to reviews:

Dear Editors,

For this final version we conducted the formatting changes as requested.

1. We removed the line numbers.
2. We removed the box/table around the list of abbreviations.
3. We changed the sentence of none declarations and "Competing interests" as suggested.
4. We added to grants to the Acknowledgments.
5. We renamed the Figure title files.
6. We cropped the figures and saved them under .jpeg.
7. The contributing authors changed the phone and fax number.

Best regards,

Andrea Staack